\documentstyle[11pt]{article}
\pagestyle{empty}


\begin{document}
\setlength{\topskip}{0cm}
\setlength{\topmargin}{0cm}
\setlength{\headsep}{0cm}
\setlength{\headheight}{0cm}
\noindent Quiz 8


\noindent Math 226, Section 02

\vskip .2in

\noindent Give the general solution of Euler's equation
$$
x^2 y'' + \alpha y' + \beta y = 0.
$$
when $\alpha = 1$ and $\beta = -1$.  Using this, either give the
particular solution satisfying $y(0) = 1$, $y'(0) = 0$, or explain
why no such solution exists.



\end{document} 
